Journal of Hepatology LONG-TERM ALBUMIN TREATMENT IN PATIENTS WITH CIRRHOSIS AND ASCITES

--Manuscript Draft--

Manuscript Number:	JHEPAT-D-21-02529R1	
Article Type:	Invited Review	
Section/Category:	Cirrhosis and Liver Failure	
Keywords:	decompensated cirrhosis; ascites; long-term albumin treatment; disease-modifying intervention; effective albumin concentration	
First Author:	Paolo Caraceni, MD	
Corresponding Author:	Paolo Caraceni, MD Alma Mater Studiorum - University of Bologna Bologna, ITALY	
Order of Authors:	Paolo Caraceni, MD	
	Alastair O'Brien	
	Pere Gines	
Manuscript Region of Origin:	ITALY	
Abstract:	Although proposed for the first time several decades ago, the possibility that long-term human albumin can be effective for treating patients with cirrhosis and ascites has become a topic of scientific and clinical discussion in the last decade. This review will critically analyze the data available highlighting the differences existing between studies, the controversial issues and the future perspectives related to such a treatment. Long-term albumin administration to patients with cirrhosis and ascites represents a completely different treatment perspective as compared to acute or short- term uses of albumin. Results from the ANSWER and the MACHT studies indicate that long-term albumin treatment can be effective, safe and able to modify the course of the disease provided that albumin is given in a sufficient amount and for a sufficient time to restore physiological levels and presumably functions of the circulating molecule, which are compromised, at least partially, in patients with decompensated cirrhosis. However, the discordant findings with other studies and several additional issues call for the critical need of further clinical studies and randomized trials to confirm the benefits of long-term albumin therapy on clinical outcomes. Other important areas for further research include the definition of the target population stratified according to the expected outcomes, biomarkers of response, the optimal dose and frequency of albumin infusions, stopping rules, and the cost-effectiveness of treatment in the different health-care systems worldwide, particularly in those where the logistic issues and costs related to periodic intravenous infusions can represent an important limitation to the implementation of this innovative approach in the clinical practice.	
Response to Reviewers:	Ref: JHEPAT-D-21-02529 Title: LONG-TERM ALBUMIN TREATMENT IN PATIENTS WITH CIRRHOSIS AND ASCITES Article Type: Invited Review Journal of Hepatology Dear Editors, we thank the reviewers and editors for their comments. We substantially agree with their criticisms and the manuscript has been changed accordingly. In particular, the length of the manuscript has been cut more than 500 words and the parts on mechanisms of action and acute uses of albumin greatly shortened. We hope that the revised manuscript will now meet your approval. Kind regards,	

On behalf of all co-authors,

Paolo Caraceni

Point by point response to the reviewers:

Reviewer #1.

This is a review about long-term albumin use in patients with cirrhosis and ascites. The Authors comprehensively reviewed and interpreted the current evidence. They also highlighted areas of uncertainties and provided suggestions for future research in this field. I have some minor comments:

1) Paragraph "Why albumin functions are useful in patients with cirrhosis and ascites". It would be better rephrasing "Why albumin infusion is useful in patients with cirrhosis and ascites".

The title of the paragraph has been rephrased. Hopefully the change will meet the reviewer's approval.

2) Page 8, non-oncotic properties of albumin. Please, mention data showing that human albumin can improve cirrhotic cardiomyopathy in experimental cirrhosis (Bortoluzzi Hepatology 2013)

A sentence describing the main finding of the study has been added.

3) Use of albumin in patients with AKI. Here it is important to highlight that albumin infusion has been mainly investigated in patients with ascites and AKI. Patients without ascites are less likely to have a severe reduction of effective circulating volume and high doses of albumin could be dangerous in this group. Moreover, a recent retrospective cohort study evaluated the use of albumin in patients with cirrhosis and AKI in US and did not find a benefit with albumin administration (Patidar KR Liver Int 2021).

Following the suggestion of the other reviewer, the paragraph on AKI has been reduced to only one sentence. We have highlighted that the scientific evidence supporting this use is still limited even if recommended by international guidelines.

4) Comments about albumin threshold to be reached to achieve clinical benefits (page 17). Here the Authors seem to suggest that a 4 g/dl threshold has to be reached to maximize the benefits of albumin. However, a note of caution is mandatory here. Physicians could be tempted to give high doses of albumin in a short time to reach the 4 g/dl threshold, but the ATTIRE trial showed that this practice could be harmful leading to circulatory overload/pulmonary edema. Therefore any albumin threshold should be reached in the mid-term (weeks) instead of short term (days). Moreover, further prospective studies are necessary to prove that this threshold can be safely achieved and can maximize albumin benefit.

We fully agree with the reviewer. The following paragraph has been added in the text: "It also emerges that the target level should be reached in weeks and not days to avoid the risk of volume overload particularly in patients who appear more prone to develop this complication due to predisposing conditions".

5) "Challenges and open issues" paragraph. Albumin proved to be effective in patients with ascites and a study comparing long term albumin administration versus other strategies (e.g., transjugular intrahepatic portosystemic shunt) would be very interesting.

We agree with the reviewer and therefore a new paragraph dealing with this issue has been added in Chapter 6: "It would be also important to perform studies comparing

long-term HA treatment with transjugular intrahepatic portosystemic shunt (TIPS). Research questions to be answered are: which are the patients in which one treatment is superior to the other? can long-term HA and TIPS be not mutually exclusive and instead part of a sequential approach for optimizing the global management of patients with ascites?" 6) Page 14 (line 49-50) ref 57 is not appropriate here. Please, check it. The reviewer is right. We have changed the number of the reference. Reviewer #2. Caraceni et al. provided a detailed review on the potential benefit of long-term benefit of human albumin (HA) in patients with cirrhosis and ascites. The review is well-written, detailed and interesting for a broad spectrum of readers in the hepatology field. They discussed adequately to two main RCT in the topic (ANSWER and MACHT) and controversies. They opened new research perspectives. Comments: 1) The acute and short-term use of human albumin in cirrhosis is outside of the scope of the review. If authors want to mention these data, they should limit it into a table. 2) The review is too long. There are already published review on albumin mechanism of action in cirrhosis. They should only mention in a figure and center their text on recent advances. We agree with both comments of the reviewer. We have therefore summarized both parts on the acute/short-term use and the mechanisms of action of albumin by cutting more than 500 words. However, we have decided to leave a separate paragraph on acute/short-term treatments, mostly dedicated to the ATTIRE study, since we believe that one of the main message of this review is to highlight the different perspective of long-term albumin treatment from the current short-term uses.



ALMA MATER STUDIORUM UNIVERSITY OF BOLOGNA DEPARTMENT OF SURGICAL AND MEDICAL SCIENCES

February 26, 2022

Prof. Paolo Angeli Editor in Chief, Journal of Hepatology

Prof. Vincent Wong Prof. Jean-Francois Dufour Prof. Robert Schwabe Prof. Norah Terrault *Guest Editors, Supplement on "Therapeutic Breakthrough in Hepatology"*

Dear Editor and Guest Editors,

on behalf of my co-authors, I am submitting the revised version of the invited review entitled "Long-term use of albumin in patients with cirrhosis and ascites" by P. Caraceni, A. O'Brien and P. Gines.

We thank the reviewers and the editor for their comments.

We have shortened the manuscript by more than 500 words and reviewed the text according to their criticisms and suggestions.

A detailed point-by-point response to the comments of the editor and reviewers has been provided.

All the listed authors have contributed actively and approved the submitted manuscript.

We hope that the manuscript will meet yours and reviewer's expectations.

Yours sincerely,

Paolo

Paolo Caraceni, M.D. Associate Professor of Internal Medicine Alma Mater Studiorum University of Bologna, Italy paolo.caraceni@unibo.it

JOURNAL OF HEPATOLOGY

Journal of Hepatology

Revised Submission Checklist

This form must be completed and submitted for all revised manuscripts. Without this form the manuscript will be returned to the corresponding author for completion.

Corresponding Author:	Paolo Caraceni
Manuscript Number:	21-02529-1

Below, provide the page number(s) or figure legend(s) where the information can be located. Please make sure that all the information requested below is present in the manuscript.

1) Submission

- a) Title page: COI, Financial support, Authors' contributions, keywords.
- b) Structured abstract and lay summary
- c) All tables and figures included, numbered correctly, with legends (p value and statistical test)
- d) Supplementary data included in a single, separate word file
- e) A detailed point by point response to reviewers comments and changes highlighted in text
- f) All authors to complete and upload an ICMJE conflict of interest form.
- g) Graphical abstract

2) Materials and methods

- a) Completed the CTAT form for all reagents and resource to be added to supplementary material
- b) Identify the source and authentication of cell lines
- c) Identify animal species, number of animals used, strain, sex and age.
- d) For animal studies include a statement of compliance with ethical regulations and identify the committee(s) approving the experiments.
- e) For qPCR data provide information according to the Minimum Information for Publication of Quantitative Real-Time PCR Experiments (MIQE) guidelines

3) Human subjects

- a) Identify the committee(s) approving the study protocol.
- **b)** Include a statement confirming that informed consent was obtained from all subjects.

Y
Y
Y
Ν
Y
Y
N

Completed, or reported on page(s) or figure legend(s):

NA	
NA	
NA	
NA	



NA	
NA	

- c) For randomized studies report the clinical trial registration number (at ClinicalTrials.gov or equivalent).
- d) For phase II and III randomized controlled trials:
 - I. Please refer to the CONSORT statement and submit the CONSORT checklist with your submission.
 - II. Include all version of the study protocol and statistical plan (to be published as supplementary information)
- e) Identify the inclusion/exclusion criteria in the selection process for the patients included in the study
- 4) Statistics
 - a) State what statistical tests were completed and why
 - **b)** Explain the sample size and how this size provides an adequate power to detect a pre-specified effect size.
- 5) Data deposition (Provide accession codes for deposited data)
 - a) When using public databases:
 - I. Identify the source and include a valid link
 - II. When using databases that require permission, include a statement confirming that permission was obtained
 - b) Data deposition in a public repository is mandatory for:
 - I. Protein, DNA and RNA sequences
 - II. Microarray data

Deposition is strongly recommended for many other datasets for which structured public repositories exist

NA			
NA	 		

NA

NA	 	
NA		

NA	
NA	

NA	
NA	
NA	

JOURNAL OF HEPATOLOGY

Ref: JHEPAT-D-21-02529 Title: LONG-TERM ALBUMIN TREATMENT IN PATIENTS WITH CIRRHOSIS AND ASCITES Article Type: Invited Review Journal of Hepatology

Dear Editors,

we thank the reviewers and editors for their comments. We substantially agree with their criticisms and the manuscript has been changed accordingly. In particular, the length of the manuscript has been cut more than 500 words and the parts on mechanisms of action and acute uses of albumin greatly shortened.

We hope that the revised manuscript will now meet your approval.

Kind regards,

On behalf of all co-authors,

Paolo Caraceni

Point by point response to the reviewers:

Reviewer #1.

This is a review about long-term albumin use in patients with cirrhosis and ascites. The Authors comprehensively reviewed and interpreted the current evidence. They also highlighted areas of uncertainties and provided suggestions for future research in this field. I have some minor comments:

1) Paragraph "Why albumin functions are useful in patients with cirrhosis and ascites". It would be better rephrasing "Why albumin infusion is useful in patients with cirrhosis and ascites".

The title of the paragraph has been rephrased. Hopefully the change will meet the reviewer's approval.

2) Page 8, non-oncotic properties of albumin. Please, mention data showing that human albumin can improve cirrhotic cardiomyopathy in experimental cirrhosis (Bortoluzzi Hepatology 2013)

A sentence describing the main finding of the study has been added.

3) Use of albumin in patients with AKI. Here it is important to highlight that albumin infusion has been mainly investigated in patients with ascites and AKI. Patients without ascites are less likely to have a severe reduction of effective circulating volume and high doses of albumin could be dangerous in this group. Moreover, a recent retrospective cohort study evaluated the use of albumin in patients with cirrhosis and AKI in US and did not find a benefit with albumin administration (Patidar KR Liver Int 2021).

Following the suggestion of the other reviewer, the paragraph on AKI has been reduced to only one sentence. We have highlighted that the scientific evidence supporting this use is still limited even if recommended by international guidelines.

4) Comments about albumin threshold to be reached to achieve clinical benefits (page 17). Here the Authors seem to suggest that a 4 g/dl threshold has to be reached to maximize the benefits of albumin. However, a

note of caution is mandatory here. Physicians could be tempted to give high doses of albumin in a short time to reach the 4 g/dl threshold, but the ATTIRE trial showed that this practice could be harmful leading to circulatory overload/pulmonary edema. Therefore any albumin threshold should be reached in the mid-term (weeks) instead of short term (days). Moreover, further prospective studies are necessary to prove that this threshold can be safely achieved and can maximize albumin benefit.

We fully agree with the reviewer. The following paragraph has been added in the text: *"It also emerges that the target level should be reached in weeks and not days to avoid the risk of volume overload particularly in patients who appear more prone to develop this complication due to predisposing conditions".*

5) "Challenges and open issues" paragraph. Albumin proved to be effective in patients with ascites and a study comparing long term albumin administration versus other strategies (e.g., transjugular intrahepatic portosystemic shunt) would be very interesting.

We agree with the reviewer and therefore a new paragraph dealing with this issue has been added in Chapter 6: *"It would be also important to perform studies comparing long-term HA treatment with transjugular intrahepatic portosystemic shunt (TIPS). Research questions to be answered are: which are the patients in which one treatment is superior to the other? can long-term HA and TIPS be not mutually exclusive and instead part of a sequential approach for optimizing the global management of patients with ascites?"*

6) Page 14 (line 49-50) ref 57 is not appropriate here. Please, check it.

The reviewer is right. We have changed the number of the reference.

Reviewer #2.

Caraceni et al. provided a detailed review on the potential benefit of long-term benefit of human albumin (HA) in patients with cirrhosis and ascites. The review is well-written, detailed and interesting for a broad spectrum of readers in the hepatology field. They discussed adequately to two main RCT in the topic (ANSWER and MACHT) and controversies. They opened new research perspectives. Comments:

1) The acute and short-term use of human albumin in cirrhosis is outside of the scope of the review. If authors want to mention these data, they should limit it into a table.

2) The review is too long. There are already published review on albumin mechanism of action in cirrhosis. They should only mention in a figure and center their text on recent advances.

We agree with both comments of the reviewer.

We have therefore summarized both parts on the acute/short-term use and the mechanisms of action of albumin by cutting more than 500 words. However, we have decided to leave a separate paragraph on acute/short-term treatments, mostly dedicated to the ATTIRE study, since we believe that one of the main message of this review is to highlight the different perspective of long-term albumin treatment from the current short-term uses.

1 2					
3					
4 5 6 7	LONG-TERM ALBUMIN TREATMENT IN PATIENTS WITH CIRRHOSIS AND ASCITES				
, 8 9 10	^{1,2*} Paolo Caraceni, ^{3,*} Alastair O'Brien, ^{4,*} Pere Gines				
11					
12 13	¹ IRCCS Azienda Ospedaliera-Universitaria di Bologna, Italy				
14 15	² Department of Medical and Surgical Sciences and Center for Biomedical Applied Research,				
16	Alma Mater Studiorum University of Bologna, Italy				
17 18	² UCL Institute for Liver and Digestive Health, Upper 3rd Floor, Division of Medicine, Royal Free Campus,				
19 20	Rowland Hill Street London NW3 2PF, UK				
21	³ Hospital Clinic of Barcelona, University of Barcelona, IDIBAPS, CIBEReHD,				
22	Barcelona, Catalonia, Spain				
24 25					
26 27	*all authors contributed equally to the manuscript				
28 29					
30 21					
32					
33 34					
35					
36 37	Manuscript word count: 5169				
38 39	Number of figures: 4				
40	Number of tables: 3				
41					
43 44					
45 46 47 48 49	Key-words: decompensated cirrhosis, ascites, long-term albumin treatment, disease-modifying intervention, effective albumin concentration.				
50 51 52	Address correspondence:				
53	Paolo Caraceni, MD				
54 55	U.O. Semeiotica Medica Department of Medical and Surgical Sciences				
56 57	Alma Mater Studiorum University of Bologna				
58	IRCCS Policlinico S. Orsola di Bologna, Italy				
59 60	Via Albertoni 15 40138 Bologna, Italy				
61	paolo.caraceni@unibo.it				
62 63					
64 65					

ABSTRACT

Although proposed for the first time several decades ago, the possibility that long-term human albumin can be effective for treating patients with cirrhosis and ascites has become a topic of scientific and clinical discussion in the last decade.

This review will critically analyze the data available highlighting the differences existing between studies, the controversial issues and the future perspectives related to such a treatment. Longterm albumin administration to patients with cirrhosis and ascites represents a completely different treatment perspective as compared to acute or short-term uses of albumin. Results from the ANSWER and the MACHT studies indicate that long-term albumin treatment can be effective, safe and able to modify the course of the disease provided that albumin is given in a sufficient amount and for a sufficient time to restore physiological levels and presumably functions of the circulating molecule, which are compromised, at least partially, in patients with decompensated cirrhosis. However, the discordant findings with other studies and several additional issues call for the critical need of further clinical studies and randomized trials to confirm the benefits of longterm albumin therapy on clinical outcomes. Other important areas for further research include the definition of the target population stratified according to the expected outcomes, biomarkers of response, the optimal dose and frequency of albumin infusions, stopping rules, and the costeffectiveness of treatment in the different health-care systems worldwide, particularly in those where the logistic issues and costs related to periodic intravenous infusions can represent an important limitation to the implementation of this innovative approach in the clinical practice.

Abstract word count: 257

- **KEY-POINTS**

 • Albumin plays an essential role in human physiology.

- Both oncotic and non-oncotic properties of the albumin molecule are likely important for antagonizing key events in the pathophysiology of decompensated cirrhosis, such as circulatory dysfunction and systemic inflammation.
- In patients with decompensated cirrhosis, circulating albumin is low; moreover, albumin is damaged and dysfunctional.
- Long-term albumin administration to patients with cirrhosis and ascites represents a completely different treatment perspective as compared to acute or short-term uses of albumin.
- Long-term albumin administration appears to be an effective and safe treatment able to modify the course of the disease. It could be speculated that treatment benefits occur when albumin is given in a sufficient amount and for a sufficient length of time to restore the physiological levels and presumably functions of the molecule.
- Additional randomized clinical trials are needed to confirm the positive effects of long-term albumin administration on clinical outcomes
- Areas for future research include the definition of the target population stratified according to the expected outcomes, the biomarkers of response, the optimal dose and frequency of albumin infusions, the stopping rules, and the cost-effectiveness of treatment in the different health-care systems worldwide.

Human albumin (HA) administration is one of the most studied interventions in patients with decompensated cirrhosis (1). Randomized clinical trials (RCTs) have produced controversial data on the efficacy and safety of HA, likely as a consequence of the great variance in terms of indications, experimental design, type of patients enrolled, length of treatment, and dosage and frequency of infusions. Current HA indications are for acute or short-term (maximum 2 weeks) administration. Although proposed for the first time several decades ago, the possibility that HA can be administered for much longer to treat patients with ascites has become a topic of scientific and clinical discussion only in the last years.

This review will critically analyze the data available, the controversial issues and the future perspectives related to long-term HA treatment, highlighting the differences existing between studies and the other short-term uses of HA administration in patients with decompensated cirrhosis.

1. The albumin molecule in patients with decompensated cirrhosis

Albumin is synthesized by hepatocytes and continuously secreted into the circulation, without being stored in the liver (2). Albumin has a concentration of 3.5-5 g/dl and accounts for more than 50% of the total circulating protein content. It has a half-life of about 20-days in healthy adults and is continuously taken up and recycled by hepatocytes (3,4).

Albumin accounts for approximately 75% of plasma oncotic pressure, due to its high concentration and net negative charge and is therefore principally responsible for fluid distribution within the body's compartments (5). Albumin also has many other biological functions termed non-oncotic properties. It reversibly binds many molecules, including drugs, metallic ions, and multiple inflammatory mediators, potentially affecting systemic inflammation, immune response, antioxidant capacity and endothelial function (3,4) (Figure 1).

In patients with decompensated cirrhosis, the albumin molecule undergoes both quantitative and qualitative changes. Hypoalbuminemia has been considered a marker of advanced liver disease for decades. It correlates with the severity of cirrhosis and independently predicts the poor outcome of these patients (6,7). Hypoalbuminemia results mainly from the reduced synthesis in the diseased liver and the enhanced catabolism due to the structural alterations of the molecule; however, the hemodilution related to the expanded total plasma volume also contributes to its reduced plasma levels (4,8,9).

Besides hypoalbuminemia, it has become evident during the last decade that the persisting inflammatory state of advanced cirrhosis induces molecular, structural and conformational changes of albumin that adversely affect its binding, transport and detoxification capacities (4,9). Albumin circulates predominantly in a reduced state with the free thiol group at the cysteine-34 (Cys-34) residue acting as a free radical scavenger for reactive oxygen and nitrogen species (3). Oxidative damage of the Cys-34, which can occur alone or in combination with other molecular changes, represents the most frequent post-translational alteration (10-12). Other structural changes include the truncation of N-terminal and C-terminal portions and the glycosylation of the molecule (11,12). As a result of these damages, albumin in decompensated cirrhosis becomes dysfunctional, showing an impairment of binding, detoxification and antioxidant activities, which parallels the severity of the disease (12,13). Damaged albumin isoforms may even be harmful since oxidized molecules have been shown to activate immune cells and promote inflammation (14,15) (Figure 1).

These findings have led researchers to propose the concept of an "effective albumin concentration", which implies that the global function of albumin, resulting from both oncotic and non-oncotic properties, is related not only to its quantitative circulating level, but also to the preservation of its structure (12,16). As damage accumulates, the proportion of the albumin

molecules maintaining a fully preserved structure declines according to the severity of the disease (11,12). Interestingly, effective albumin concentration appears to discriminate the different stages of cirrhosis and predict outcomes significantly better than the total serum albumin concentration routinely measured by standard laboratory methods in daily clinical practice (12).

The combination of low circulating concentration of albumin and its dysfunctional quality observed in patients with decompensated cirrhosis provides the rationale for exogenous HA infusions aiming to restore the major physiological functions of the molecule.

2. Pathophysiological rationale for the use of albumin in patients with decompensated cirrhosis

Studies on the use of HA infusions were first reported more than 70 years ago (17), and all international guidelines currently recommend HA as the fluid of choice for volume expansion in patients with cirrhosis and ascites (18-19). Indeed, HA has been consistently shown to improve effective hypovolemia, reduce the activity of vasoconstrictor systems and increase mean arterial pressure in patients undergoing large-volume paracentesis (LVP) or suffering spontaneous bacterial peritonitis (SBP) or hepatorenal syndrome (HRS) (20-23).

Clinical and experimental data have raised the possibility of HA infusions having other beneficial properties beyond volume expansion (4,16,24). HA administration in patients with SBP improves systemic hemodynamics through mechanisms not directly related to volume expansion, but consistent with improvement of endothelial function (25,26). HA also appears to improve cardiac contractility in an experimental model of isolated perfused rat heart by reducing the activation of inflammatory mediators in the cardiac tissue (27). Furthermore, cirrhosis associated prostaglandin E2-mediated immune dysfunction was improved following HA infusion (28,29) and analyses of samples from two trials in cirrhosis patients demonstrated that HA is able to reduce systemic

inflammation (25,30). Finally, recent experimental evidence indicates that HA internalizes in immune cells and modulates their responses through interaction with endosomal toll-like receptor signalling (31).

Therefore, the potential benefits resulting from both oncotic and non-oncotic properties of the molecule provide the rationale for exogeneous HA infusions aiming to counteract the two major pathogenic drivers of poor outcome in patients with decompensated cirrhosis, namely effective hypovolemia and systemic inflammation/immune dysfunction (32) (Figure 2).

3. Long-term albumin treatment in patients with ascites

Five to ten percent of patients with compensated cirrhosis develop ascites every year, which represents the most frequent decompensating event of cirrhosis (1,33-35). Moderate and massive (grade 2 and 3) ascites usually requires long-term treatment, leads to recurrent hospitalizations, also caused by related complications (i.e., SBP, HRS, abdominal hernias, and restrictive ventilatory dysfunction), and impairs patient quality of life (36). Therefore, the contribution of ascites to the heavy health economic burden of decompensated cirrhosis is highly relevant (37). Finally, the development of ascites has a dramatic negative impact on patient prognosis, as the 1, 2 and 5-year mortality is approximately 30%, 50%, and 70%, respectively (34).

Based on its oncotic activity, chronic use of HA to treat ascites was proposed many decades ago, but the studies, uncontrolled and/or very small sized, failed to show a clear benefit (38,39). After almost 40 years, two RCTs, enrolling a relatively small number of patients, showed a better response of ascites to HA in addition to diuretics during hospitalization (40) and a significantly lower recurrence of grade 2-3 ascites associated with higher transplant-free survival in the group receiving long-term HA for a median follow-up of 84 months (41). More than 10 years after these pivotal studies, three clinical trials have been published in 2018 (42-44).

3.1 The ANSWER trial

The ANSWER study (42), a non-profit, Italian multicenter, open-label, pragmatic RCT, enrolled 431 patients with persisting non complicated grade 2 and 3 ascites requiring the combination of an anti-mineralocorticoid drug (at least 200 mg/day) and furosemide (at least 25 mg/day) to receive either standard medical treatment (SMT) or SMT plus 40 g of HA twice a week for the initial 2 weeks and then 40 g once a week.

HA administration improved the management of ascites, as the need of large-volume paracentesis and the incidence of refractory ascites decreased by about 50%. The incidence rate of other complications (i.e., SBP, non-SBP bacterial infections, hepatic encephalopathy (HE) grade III or IV, HRS type 1, renal dysfunction [serum creatinine >1.5 mg/dl], moderate hyponatremia or hyperkalemia) decreased also by 30-67%, leading to lower liver-related hospitalizations and days spent in hospital per year as compared to the control group, which were reduced by 35 and 45% respectively. A significantly better 18-month overall survival, which was the primary end-point of the study, was observed in patients receiving albumin, with a 38% reduction of the hazard ratio for mortality. The multivariable risks analysis for 18-month all-cause mortality considering TIPS placement or liver transplantation as competing events showed that HA treatment was the sole variable associated with increased survival. Furthermore, patients receiving HA had a better quality of life and HA treatment proved to be also cost-effective based on the reimbursement rates from the Italian National Health Service as compared to standard therapy. Finally, sideeffects were similar between the two groups and no episodes of volume overload related to HA occurred, and only very few mild allergic reactions to HA infusion were reported

A post-hoc analysis has highlighted the importance of serum albumin concentration in the interpretation of the positive results of the study (45). With the schedule of HA administration followed in the ANSWER trial, serum albumin concentration increased from a median level of 3.1

g/dl to almost 4 g/dl after 1 month of treatment, and remained stable afterwards. In the control group, no rise above the baseline level was instead observed, so that the difference between the two groups (about 0.7-0.8 g/dl) was significant during the entire follow-up.

Furthermore, in patients undergoing long-term HA serum albumin concentration at 1 month of treatment, but not baseline serum albumin, directly correlated with the probability of 18-month overall survival, with the best discriminating cut-off to independently predict survival identified at 4 g/dl. The two baseline factors that independently predicted the achievement of this cut-off were serum albumin concentration and MELD score, so that the lower the baseline serum albumin or higher the MELD score, the lower the probability of reaching the threshold of serum albumin shown above (45).

The major limitation of the ANSWER study was of course related to its open label design. Although the absence of blinding reduces the internal validity of the study, the pragmatic design of the ANSWER trial could have produced an even higher external validity since other strengths, such as large sample-size, prolonged follow-up, and a hard primary endpoint, are satisfied. More important than blinding is the fact that weekly HA infusions led patients to be seen more frequently by health care professionals than those enrolled in the control group. Although patients in the ANSWER trial were usually not evaluated by physicians during the HA infusions and some of them received HA in residential services or even at home, it cannot be excluded *a priori* that the regular contacts with health care services and personnel may have produced a better general management, thus contributing to the improved outcomes. Again, the real-word assessment of the entire pathway of care related to the intervention under study is a core feature of pragmatic trials. It is also tempting to speculate that the need for regular intravenous infusions, if perceived by patients as beneficial to their health, could favor the adherence of such a challenging group of patients to their overall pathway of care and incentivize them to overcome logistic limitations when present.

Finally, almost half of patients included in the ANSWER study had cirrhosis caused by hepatitis C, which remained untreated during the study, and patients with active alcoholism were not included. Thus, the effects of HA administration in those with alcohol-related cirrhosis and active drinking remains to be determined.

3.2 The "refractory ascites trial"

The core results of the ANSWER trial were confirmed by a prospective, non-randomized clinical trial performed in Padua, Italy, which enrolled 70 patients with cirrhosis and refractory ascites (43). Patients who received SMT + HA (20 g twice a week) had a significantly lower 24-month mortality than the 25 patients receiving the SMT. Treatment with HA was the sole independent protective factor of death and it was associated with a significantly lower cumulative incidence of re-hospitalizations due to HE, accumulation of ascites, and bacterial infections.

3.3 The MACHT trial

The midodrine and albumin for prevention of complications in patients with cirrhosis awaiting liver transplantation (MACHT) trial was a ground-breaking trial for two reasons (44). First, it is one of the first RCTs in patients with decompensated cirrhosis to evaluate the effect of combination therapy, midodrine (an alpha-adrenergic agonist) and HA, to improve outcomes of patients with cirrhosis; and second, it used a primary end-point that combines the most relevant complications of advanced cirrhosis (renal failure, hyponatremia, infections, HE and gastrointestinal bleeding). Previous RCTs in cirrhosis had evaluated a single therapy, usually a drug, to treat or prevent a single complication of the disease (46). The rationale for the use of such combination of vasoconstrictor drug and HA was to normalize the impaired effective arterial blood volume thought to be responsible, at least in part, for some complications of cirrhosis (22,47,48). This specific treatment combination was based on the positive results of midodrine together with HA in the treatment of HRS (49)

Unfortunately, the MACHT trial did not meet the primary endpoint and therapy with midodrine and HA was not associated with reduction in the incidence of complications or mortality in patients with decompensated cirrhosis awaiting liver transplantation. The only positive effect found was a moderate suppression of activity of renin-angiotensin and sympathetic nervous systems, suggesting a beneficial effect on circulatory function.

Speculation about possible causes of lack of efficacy of combined therapy in MACHT trial is challenging. One possibility is that midodrine, which has a relatively weak vasoconstrictor potency compared to that of terlipressin (22,50) did not produce a sufficient vasoconstriction in the splanchnic arteries and therefore was not capable of improving effective arterial blood volume sufficiently. Alternatively, but not mutually exclusive, the dose of HA used in the study (40 g every 2 weeks) could have been insufficient to produce the expected hemodynamic and nonhemodynamic effects of HA (25,32). In the treated group, serum albumin concentration increased from a mean of 30 g/L at baseline to a mean of 34 g/L at week 12. However, a similar increase was observed in patients from the placebo group, suggesting that the improvement was unrelated to therapy. Another possibility is that the duration of treatment was not long enough to cause a significant beneficial effect on patients. In fact, because many patients were transplanted quite rapidly during the study, the mean duration of treatment in the midodrine and albumin group was of less than 3 months (average 80 days). Finally, it is also theoretically possible that the hypothesis of the study was wrong. However, the hypothesis cannot be ruled out completely because the combined treatment fell short of achieving a normalization of effective arterial blood volume.

4. Acute and short-term treatment in patients with decompensated cirrhosis

HA treatment given from a single administration to a short-term course up to a maximum of 15 days, with the purpose of preventing or treating acute complications of decompensated cirrhosis, has been the main, if not the sole, use of HA for the last 40 years.

International guidelines consistently recommend HA infusion to prevent circulatory dysfunction after paracentesis and renal failure in patients with SBP or to diagnose and treat hepatorenal syndrome (HRS) (18,19). HA is also recommended by international guidelines in algorithms of diagnosis and management of acute kidney injury (AKI) and hypervolemic hyponatremia although solid evidence from clinical studies is still limited (18,19,51,52). In contrast, no benefits in survival were observed in RCTs assessing HA administration in patients with infections other than SBP (30,53,54). Noteworthy, HA infusion significantly increased the risk of pulmonary edema in one of these studies (54). Negative results were also observed in patients presenting with an acute episode of hepatic encephalopathy (55).

Finally, the results of a large multicenter, open label, randomized, controlled trial assessing shortterm HA administration to prevent complications in hospitalized patients have been recently published (56). In the ATTIRE (Albumin to Prevent Infection in Chronic Liver Failure) study, 777 patients with cirrhosis hospitalized with a decompensating even were included in the analysis. The treatment arm received daily intravenous infusions of HA to increase and maintain a serum albumin level of at least 3.0 g per liter throughout the trial treatment period of up to 14 days. The standard-care (control) group received HA only as recommended by guidelines (18,19), after largevolume paracentesis, during SBP, or for HRS. There were no significant differences in the composite primary end-point (infection, renal failure, or death) between treatment and control groups, despite the HA group patients receiving a significantly much higher HA dose compared to those in standard care. The HA group also had more severe or life-threatening serious adverse events, especially pulmonary oedema or fluid overload. The trial's greatest strength was the large number of patients recruited and multiple sites involved. Other strengths were that patients were recruited very soon after hospitalization (on average one day) and sub-group analyses including a time to event and threshold analysis that included missing data. Finally, crucial for an open-label trial ATTIRE achieved substantial differences in the amounts of 20% HA infused between the albumin treated group and control patients especially during the early part of the trial when an increased albumin level might have been expected to have benefit.

ATTIRE had several limitations, most obviously it was not blinded. 90% of recruited patients had alcohol-induced cirrhosis and results may differ for other causes of liver disease. The components of the composite end point (infection, renal dysfunction, and mortality) were not equivalent in severity; however, these do represent a common disease trajectory and move in line with each other and the 3 and 6-month mortality outcomes were also null. Inevitably in a trial of this size there was heterogeneity in patients recruited in terms of infection, antibiotic treatment and organ dysfunction and it is possible that a specific group of patients might have benefited that were not examined in the subgroup analyses. However, no biomarkers or clinical features that predict potential benefit from albumin infusions have been identified to enable such an approach to date. Thus, the results of this study do not support the use of HA in patients admitted for worsening or onset of a complication of cirrhosis with the aim of preventing the development of further complications of the disease during hospitalization.

5. Long-term albumin treatment: what we have learnt so far?

It appears evident that the data on efficacy and safety of HA are quite heterogeneous, so that some clinicians and researchers emphasize the benefits of HA in many conditions of

decompensated cirrhosis, whereas others are against any extension of its use beyond the few well-established indications.

A strong scientific debate is currently ongoing regards long-term HA administration in patients with ascites, which represents not only a novel indication, but also carries logistic and economic issues inherent to the need for chronic periodic intravenous infusions that render its applicability in clinical practice problematic.

It would be foolish to propose long-term HA administration for all patients with ascites who can be quite different in terms of clinical phenotypes and prognosis. As for any other intervention, the objective should be to define the target sub-group populations for whom the benefits are significant, the modalities of administration in terms of dosage, frequency and length of treatment, the assessment of response, and the absolute and relative contraindications.

At present, the clinical phenotype of cirrhosis who can benefit from long-term HA treatment appears to be represented mostly by patients with relatively stable conditions and at least grade 2 non complicated ascites despite a moderate dosage of diuretics. Patients who had recently resolved an acute complication of the disease yet still presenting with ascites are also amenable to treatment. For these types of patients, administration of long-term HA has been recently included among the medical treatment options for the management of ascites by the Italian Association for the Study of the Liver (AISF) (57). These recommendations suggest the use of long-term HA also in refractory ascites, because by adding HA some of the patients may become responsive to diuretics (43). It should be acknowledged, however, that Italy represents a sort of "unicum" with respect to other countries since long-term HA treatment is reimbursed by the National Health System and is currently standard of care in many hepatological centers.

A second important issue regards the modalities of treatment. At present, the data available indicate that two conditions, strictly interrelated, need to be achieved in order to optimize long-

term HA treatment: administering enough HA to have an impact on serum albumin concentration and for enough time to unveil the clinical benefits.

The first assumption is based on the comparison between the ANSWER and the MACHT trials (42,44) (Figure 3). While HA administration in the MACHT trial did not significantly influence serum albumin concentration, which remained almost identical to that of controls, this was not the case of the ANSWER patients who present a significant increase up to almost a median of 4 g/dl after 1 month of treatment to remain thereafter significantly higher than controls throughout all the 18 months of follow-up. These divergent findings were likely due to the lower amount of HA infused in the MACHT trial, which was less than half of the amount received by the ANSWER patients.

If increasing serum albumin concentration is needed for the therapy to be effective, the question arises on which is the target level to achieve during treatment. A post-hoc analysis of the ANSWER database provides interesting information to clarify this issue (45). First, the percentage of patients with normal serum albumin concentration (>3.5 g/dl) increased from the baseline 25% to almost 80% after 1 month of treatment. Second, serum albumin concentration reached after 1 month of treatment independently and directly correlated with the 18-month survival. Third, the best discriminating cut-off level of serum albumin concentration between patients receiving or not HA was 4 g/dl.

Thus, it appears that normalizing serum albumin concentration should be the target to obtain good clinical outcomes and a maximal benefit is reached with levels around 4 g/dl. Other observations support this assumption. The pilot PRECIOSA trial, a pathophysiological study comparing the effects of high (1.5 g/kg every week) versus low (1 g/kg every 2 weeks) doses of HA given for 12 weeks in patients with decompensated cirrhosis and severe circulatory dysfunction, showed that the low dose protocol produced a significant increase of serum albumin level concentration without reaching the normal range, while the high dose protocol was able to

normalize serum albumin in all patients with a median level close to 4 g/dl (25). Interestingly, only patients receiving high doses of HA presented a significant improvement of cardiocirculatory dysfunction and systemic inflammation (25). Furthermore, even though the normal lower limit of serum albumin concentration has been set at 3.5 g/dl, more than 90% of healthy subjects of ages up to 80 years old present a serum albumin concentration higher, often quite higher, than 4 g/dl (58). Thus, the real physiological level of albumin in healthy individuals is at least around 4 g/dl.

The second assumption is that the benefits of long-term HA become evident after weeks of treatment, usually 1-2 months. Interestingly, in the ANSWER trial, the Kaplan-Meier curves of survival and refractory ascites (and also those of several other secondary end-points [Caraceni, personal communication]) started to diverge after 1-2 months of treatment once the increase in the albumin concentration had occurred and stabilized. The negative results of the ATTIRE trials provide an indirect confirmation to this hypothesis. The median length of HA treatment in the ATTIRE patients was only 8 days and, although the individualized protocol of administration significantly increased the very low baseline median serum albumin concentration reached a level little above 3 gr/dl throughout the entire 14-day follow-up (56). Along with these reasoning, it could be proposed the need of higher doses of HA in these very sick patients, which, however, would likely lead to an unacceptable risk of pulmonary edema if given in a short timeframe.

Thus, it can be concluded that long-term HA represents a completely different treatment paradigm compared to all other acute or short-term uses (Table 1). Acute or short-term treatment can last one or more days, but no longer than 2 weeks. They are usually applied in hospitalized patients (except in some of the patients subjected to after LVP or presenting AKI), either in regular wards or ICUs, but often very sick, with the goal of treating or preventing a specific acute complication. As treatment needs to become rapidly effective, high amounts of HA are often

infused in a relatively short time, thus raising safety issues related to volume overload at least in some complications, such as non-SBP bacterial infections (54) or HRS (59), and in patients admitted to hospital for an acute decompensation of the disease, as documented in the ATTIRE trial (56).

In contrast, long-term HA treatment lasts at least weeks, usually months or even years; it is usually initiated in relatively stable outpatients, but it can be started in hospitalized patients once complications are resolved, with the goal of controlling ascites and preventing other complications, thus modifying the course of the disease. HA doses are lower than those for acute indications and are distributed over a much longer time, thus making treatment safe. In this regard, no cases of volume overload and pulmonary edema have been described in the clinical trials assessing long-term administration (42-44), in contrast to the significantly higher incidence reported in some of the studies evaluating short-term HA treatment (54,56,59). The ATTIRE and ANSWER trials are examples of short- and long-term HA treatment, respectively (Table 2).

Based on all these considerations, it could be hypothesized that the goal of long-term HA administration should be to restore the physiological functions of albumin (both oncotic and non-oncotic properties), which are active against effective hypovolemia and systemic inflammation and are instead partially lost in patients with cirrhosis with ascites (4,9,32). In practical terms, the goal of long-term treatment should be filling the gap existing between the baseline patient serum albumin concentration and the on-treatment target serum albumin concentration corresponding to the physiological levels observed in healthy individuals (58) (Figure 4). As the extent of this gap is variable depending mostly on the starting level of serum albumin and on the severity of cirrhosis (45), the need emerges to go beyond a fixed dosage and schedule of HA administration - as used in the ANSWER study - to a more individualized treatment. It also emerges that the target level

should be reached in weeks and not days to avoid the risk of volume overload particularly in patients who appear more prone to develop this complication due to predisposing conditions.

In this perspective, the time-course changes of serum albumin concentration together with the clinical response in controlling ascites could be used as a guide to maximize the benefits of treatment, define stopping rules and optimize HA utilization. Further investigations are warranted to support this hypothesis.

6. Challenges and open issues of long-term albumin treatment

A number of important questions remain to be answered in clinical research on long-term HA administration in decompensated cirrhosis (Table 3).

Investigation on HA is hampered by the lack of an objective biomarker of the effect of therapy. This is in turn responsible, at least in part, for the lack of dose-finding trials. In fact, all RCTs evaluating the efficacy of HA in different indications were performed using arbitrary doses, except for the ATTIRE trial (56), in which a preliminary study was performed to assess the dose required to increase serum albumin levels above 3 g/dL (60).

Which candidate biomarker should be used to assess response to therapy? Should this biomarker be serum albumin levels? In this regard, the post-hoc analysis of the ANSWER trial showed that normalization of serum albumin concentration at one month of therapy was associated with better outcomes and increasing serum albumin levels correlated with higher survival rates (45). However, the possible usefulness of effective albumin concentration reflecting the portion of the albumin molecule pool with normal structure and function merits also to be explored (12,61). Or should this biomarker be related to some desirable beneficial effects on systemic inflammation, circulatory or liver function, such as C-reactive protein, fatty-acid binding proteins, copeptin or cell death markers, to cite a few (10,62-65). The availability of biomarkers in clinical practice would also help physicians to answer some other open questions related to long-term treatment: what is the minimum dose of HA and intervals between infusions that are effective? which are the stopping rules? when patients no longer need HA due to the improvement of their clinical conditions or treatment becomes instead futile? Another clinical issue requiring further investigation regards the more precise definition of the target population that can benefit from long-term HA therapy. Comparison of patients from ANSWER and MACHT trials indicate that patients from the former trial had less advanced cirrhosis as indicated by lower median MELD scores (12-13 vs 17-16, respectively). Therefore, the possibility exists that HA is less efficacious in patients with more severe liver insufficiency. This possibility is intriguing and deserves investigation focused on the mechanism(s) that may be responsible for this, should this hypothesis be correct.

It would be also important to perform studies comparing long-term HA treatment with transjugular intrahepatic portosystemic shunt (TIPS). Research questions to be answered are: which are the patients in which one treatment is superior to the other? can long-term HA and TIPS be not mutually exclusive and instead part of a sequential approach for optimizing the global management of patients with ascites?

Other aspects of albumin therapy also deserve attention. Given the high cost and low availability of albumin, particularly in developing countries, the issue of cost-effectiveness is very important. Results from the ANSWER trial show that therapy is cost-effective because the extra-cost of HA administration is compensated for by the decrease in hospital readmissions related to prevention of cirrhosis complications (42). Specific analyses performed in other areas of the world including direct and indirect costs are needed to have the full picture of the cost-effectiveness of long-term HA treatment worldwide. Finally, weekly HA infusions lasting about 30-60 minutes may also cause significant logistic problems related to availability of space, journey of patients from home to

hospital, availability of nurses to perform intravenous infusions, and time required for treatment. All these issues have to be taken into account should long-term HA treatment be implemented for patients with decompensated cirrhosis.

7. Conclusion

Results from the ANSWER trial represent an important step forward in the investigation of albumin as therapeutic agent for patients with decompensated cirrhosis. Besides controlling ascites, long-term HA treatment appears to significantly prevent complications and hospitalizations and improve survival, thus representing one of the candidates for being the first disease-modifying pathophysiological intervention in patients with cirrhosis and ascites. However, the discordant findings with other studies and several open issues call for the critical need of further clinical studies and randomized trials. In this regard, a large multicenter open-label RCT assessing the "effects of long-term administration of human albumin on subjects with decompensated cirrhosis and ascites" (PRECIOSA study; NCT03451292) is underway and results are eagerly awaited, even if a double-blind design would have increased the strength of the findings.

AKNOWLEDGEMENTS

We thank Dr. Maurizio Baldassarre (University of Bologna, Italy) for the contribution given to Figures.

FINANCIAL SUPPORT

Paolo Caraceni: the authors' research is funded and supported by the EU Horizon 2020 Programmes (Carbalive, Grant/Award Number: 634579; LiverHope, Grant/Award Number: 731875; Decision, Grant/Award Number: 847949).

Alastair O'Brien: the author's research is funded and supported by the National Institute for

Health Research Health Technology Assessment grant awarded, ref No. 17/67/01.

Pere Ginès: the authors' research is funded and supported by the Instituto de Salud Carlos III through the Plan Estatal de Investigación Científica y Técnica y de Innovación (Pl 16/00043 and Pl20/00579), cofunded by the European Regional Development Fund (FEDER), EU Horizon 20/20 Programme (LiverHope, Grant/Award Number: 731875) and from the Agencia de Gestió d'Ajuts Universitaris i de Recerca (AGAUR– 2017- SGR -01281).

CONFLICTS OF INTEREST

Dr. Caraceni reports research grants from Grifols SA and Octapharma SA and personal fees from Grifols SA, Kedrion Biopharma SpA, CSL Behring SA and Biotest SA. *Dr. O'Brien* reports no conflicts of interest. *Dr. Ginès* reports grants from Grifols and personal fees from CSL Behring.

AUTHORS' CONTRIBUTIONS

All authors contributed equally to the manuscript and approved the final draft for submission

REFERENCES

- 1. Ginès P, Krag A, Abraldes JG, Solà E, Fabrellas N, Kamath PS. Liver cirrhosis. Lancet 2021;398:1359-1376.
- 2. Strauss AW, Donohue AM, Bennett CD, Rodkey JA, Alberts AW. Rat liver preproalbumin: in vitro synthesis and partial amino acid sequence. Proc Natl Acad Sci USA 1977;74:1358-1362.
- 3. Fanali G, Di Masi A, Trezza V, Marino M, Fasano M, Ascenzi P. Human serum albumin: From bench to bedside. Mol Aspects Med 2012;33:209-290.
- 4. Bernardi M, Angeli P, Claria J, Moreau R, Gines P, Jalan R, et al. Albumin in decompensated cirrhosis: new concepts and perspectives. Gut 2020;69:1127-1138.
- 5. Lundsgaard-Hansen P. Physiology and pathophysiology of colloid osmotic pressure and albumin metabolism Curr Stud Hematol Blood Transfus 1986;53:1-17
- 6. Llach J, Ginès P, Arroyo V, Rimola A, Titó L, Badalamenti S, at al. Prognostic value of arterial pressure, endogenous vasoactive systems, and renal function in cirrhotic patients admitted to the hospital for the treatment of ascites. Gastroenterology 1988;94:482-7.
- 7. D'Amico G, Maruzzelli L, Airoldi A, Petridis I, Tosetti G, Rampoldi A, D'Amico M, et al. Performance of the model for end-stage liver disease score for mortality prediction and the potential role of etiology. J Hepatol 2021;75:1355-1366.
- 8. Arques S, Ambrosi P. Human serum albumin in the clinical syndrome of heart failure. J Card Fail. 2011 Jun;17(6):451-8.
- 9. Naldi M, Baldassarre M, Domenicali M, Bartolini M, Caraceni P. Structural and functional integrity of human serum albumin: Analytical approaches and clinical relevance in patients with liver cirrhosis. J Pharm Biomed Anal 2017;144:138-153.

10. Clària J, Stauber RE, Coenraad MJ, Moreau R, Jalan R, Pavesi M, et al. Systemic inflammation in decompensated cirrhosis. Characterization and role in acute-on-chronic liver failure. Hepatology 2016;64:1249-1264.

- 11. **Domenicali M, Baldassarre M,** Giannone FA, Naldi M, Mastroroberto M, Biselli M, et al. Posttranscriptional changes of serum albumin: Clinical and prognostic significance in hospitalized patients with cirrhosis. Hepatology 2014;60:1851-1860.
- 12. Baldassarre M, Naldi M, Zaccherini G, Bartoletti M, Antognoli A, Laggetta M, et al. Determination of effective albumin in patients with decompensated cirrhosis: Clinical and prognostic implications. Hepatology 2021;74:2058-2073.
- 13. Jalan R, Schnurr K, Mookerjee RP, Sen S, Cheshire L, Hodges S, et al. Alterations in the functional capacity of albumin in patients with decompensated cirrhosis is associated with increased mortality. Hepatology 2009;50:555-564.
- 14. Das S, Maras JS, Hussain MS, Sharma S, David P, Sukriti S, et al. Hyperoxidized albumin modulates neutrophils to induce oxidative stress and inflammation in severe alcoholic hepatitis. Hepatology 2017;65:631-646.
- 15. Alcaraz-Quiles J, Casulleras M, Oettl K, Titos E, Flores-Costa R, Duran-Güell M, et al. Oxidized albumin triggers a cytokine storm in leukocytes through p38 mitogen-activated protein kinase: role in systemic inflammation in decompensated cirrhosis. Hepatology 2018;68:1937-1952.
- 16. Garcia-Martinez R, Caraceni P, Bernardi M, Gines P, Arroyo V, Jalan R. Albumin: Pathophysiologic basis of its role in the treatment of cirrhosis and its complications. Hepatology 2013;58:1836-1846.
- 17. Kunkel HG, Labby DH, Ahrens EH Jr, Shank RE, Hoagland CL. The use of concentrated human serum albumin in the treatment of cirrhosis of the liver. J Clin Invest 1948;27:305-319.

 18. European Association for the Study of the Liver. EASL clinical practice guidelines for the management of patients wit decompensated cirrhosis. J Hepatol 2018;69:406-460.

- Biggins SW, Angeli P, Garcia-Tsao G, Ginès P, Ling SC, Nadim MK, et al. Diagnosis, evaluation, and management of ascites, spontaneous bacterial peritonitis and hepatorenal syndrome: 2021 Practice Guidance by the American Association for the Study of Liver Diseases. Hepatology. 2021;74:1014-1048.
- 20. Ginès A, Fernández-Esparrach G, Monescillo A, Vila C, Domènech E, Abecasis R, et al. Randomized trial comparing albumin, dextran 70, and polygeline in cirrhotic patients with ascites treated by paracentesis. Gastroenterology 1996;111:1002-1010.
- 21. Ginès P, Titó L, Arroyo V, Planas R, Panés J, Viver J, et al. Randomized comparative study of therapeutic paracentesis with and without intravenous albumin in cirrhosis. Gastroenterology 1988; 94:1493-1502.
- 22. Ginès P, Solà E, Angeli P, Wong F, Nadim MK, Kamath PS. Hepatorenal syndrome. Nat Rev Dis Primers 2018; 4:23.
- 23. Ortega R, Ginès P, Uriz J, Cárdenas A, Calahorra B, De Las Heras D, et al. Terlipressin therapy with and without albumin for patients with hepatorenal syndrome: Results of a prospective, nonrandomized study. Hepatology 2002;36:941-8.
- 24. Arroyo V, García-Martinez R, Salvatella X. Human serum albumin, systemic inflammation, and cirrhosis. J Hepatol 2014;6:396-407.
- 25. Fernández J, Clària J, Amorós A, Aguilar F, Castro M, Casulleras M, et al. Effects of albumin treatment on systemic and portal hemodynamics and systemic inflammation in patients with decompensated cirrhosis. Gastroenterology 2019;157:149-162.

- 26. Fernández J, Monteagudo J, Bargallo X, Jiménez W, Bosch J, Arroyo V, et al. A randomized unblinded pilot study comparing albumin versus hydroxyethyl starch in spontaneous bacterial peritonitis. Hepatology 2005; 42:627-634.
- 27. Bortoluzzi A, Ceolotto G, Gola E, Sticca A, Bova S, Morando F, et al. Positive cardiac inotropic effect of albumin infusion in rodents with cirrhosis and ascites: Molecular mechanisms. Hepatology 2013;57:266-76.
- 28. O'Brien AJ, Fullerton JN, Massey K a, Auld G, Sewell G, James S, et al. Immunosuppression in acutely decompensated cirrhosis is mediated by prostaglandin E2. Nat Med 2014;20:1-8.
- 29. China L, Maini A, Skene SS, Shabir Z, Sylvestre Y, Colas RA, et al. Albumin counteracts immune-suppressive effects of lipid mediators in patients with advanced liver disease. Clin Gastroenterol Hepatol 2018;16:738-747.
- 30. Fernández J, Angeli P, Trebicka J, Merli M, Gustot T, Alessandria C, et al. Efficacy of albumin treatment for patients with cirrhosis and infections unrelated to spontaneous bacterial peritonitis. Clin Gastroenterol Hepatol 2019;18:963-973.
- 31. Casulleras M, Flores-Costa R, Duran-Güell M, Alcaraz-Quiles J, Sanz S, Titos E, et al. Albumin internalizes and inhibits endosomal TLR signaling in leukocytes from patients with decompensated cirrhosis. Sci Transl Med. 2020;12(566): eaax5135
- 32. **Tufoni M, Baldassarre M**, Zaccherini G, Antognoli A, Caraceni P. Hemodynamic and systemic effects of albumin in patients with advanced liver disease. Curr Hepatol Rep 2020;Jul 1:1-12.
- 33. Ginés P, Quintero E, Arroyo V, Terés J, Bruguera M, Rimola A, et al. Compensated cirrhosis: natural history and prognostic factors. Hepatology 1987;7:122-8.
- 34. D'Amico G, Garcia-Tsao G, Pagliaro L. Natural history and prognostic indicators of survival in cirrhosis: A systematic review of 118 studies. J Hepatol 2006;44:217-231.

- б
- 35. Villanueva C, Albillos A, Genescà J, Garcia-Pagan JC, Brujats A, Calleja JL, et al. Bacterial infections adversely influence the risk of decompensation and survival in compensated cirrhosis. J Hepatol 2021;75:589-599.
- 36. Solà E, Watson H, Graupera I, Turón F, Barreto R, Rodríguez E, et al. Factors related to quality of life in patients with cirrhosis and ascites: relevance of serum sodium concentration and leg edema. J Hepatol 2012;57:1199-1206.
- 37. Ma C, Qian AS, Nguyen NH, Stukalin I, Congly SE, Shaheen AA, et al. Trends in the economic burden of chronic liver diseases and cirrhosis in the United States: 1996-2016. Am J Gastroenterol 2021;116:2060-2067.
- Wilkinson P, Sherlock S. The effect of repeated albumin infusions in patients with cirrhosis.
 Lancet 1962;2:1125-1129.
- 39. Losowsky MS, Atkinson M. Intravenous albumin in the treatment of diuretic-resistant ascites in portal cirrhosis. Lancet 1961;2:386-389.
- 40. Gentilini P, Casini-Raggi V, Di Fiore G, Romanelli RG, Buzzelli G, Pinzani M, et al. Albumin improves the response to diuretics in patients with cirrhosis and ascites: results of a randomized, controlled trial. J Hepatol 1999;30:639-645.
- 41. Romanelli RG, La Villa G, Barletta G, Vizzutti F, Lanini F, Arena U, et al. Long-term albumin infusion improves survival in patients with cirrhosis and ascites: an unblinded randomized trial. World J Gastroenterol 2006;12:1403-1407.
- 42. Caraceni P, Riggio O, Angeli P, Alessandria C, Neri S, Foschi FG, et al. Long-term albumin administration in decompensated cirrhosis (ANSWER): an open-label randomised trial. Lancet 2018;391:2417-2429.

 43. Di Pascoli M, Fasolato S, Piano S, Bolognesi M, Angeli P. Long-term administration of human albumin improves survival in patients with cirrhosis and refractory ascites. Liver Int 2019;39:98-105.

- 44. Solà E, Solé C, Simón-Talero M, Martín-Llahí M, Castellote J, Garcia-Martínez R, et al. Midodrine and albumin for prevention of complications in patients with cirrhosis awaiting liver transplantation. A randomized placebo-controlled trial. J Hepatol 2018;69:1250-1259.
- 45. **Caraceni P, Tufoni M,** Zaccherini G, Riggio O, Angeli P, Alessandria C, et al. On-treatment serum albumin level can guide long-term treatment in patients with cirrhosis and uncomplicated ascites. J Hepatol 2021;74:340-349.
- 46. **Solà E, Pose E,** Campion D, Piano S, Roux O, Simon-Talero M et al. Endpoints and design of clinical trials in patients with decompensated cirrhosis: Position paper of the LiverHope Consortium. J Hepatol 2021;74:200-219.

47. Ginès P and Schrier RW. Renal failure in cirrhosis. N Engl J Med 2009; 361:1279-1290.

- 48. Simonetto DA, Gines P, Kamath PS. Hepatorenal syndrome: pathophysiology, diagnosis, and management. Brit Med J. 2020;370:m2687.
- 49. Angeli P, Volpin R, Gerunda G, Craighero R, Roner P, Merenda R, et al. Reversal of type 1 hepatorenal syndrome with the administration of midodrine and octreotide. Hepatology 1999;29:1690-1697.
- 50. Cavallin M, Kamath PS, Merli M, Fasolato S, Toniutto P, Salerno F, et al. Terlipressin plus albumin versus midodrine and octreotide plus albumin in the treatment of hepatorenal syndrome: A randomized trial. Hepatology 2015;62:567-574.
- 51. Angeli P, Ginès P, Wong F, Bernardi M, Boyer TD, Gerbes A, et al. Diagnosis and management of acute kidney injury in patients with cirrhosis: revised consensus recommendations of the International Club of Ascites. J Hepatol 2015;62:968-974.

- 52. De Franchis R, Bosch J, Garcia-Tsao G, Reiberger T, Ripoll C; Baveno VII Faculty. Baveno VII Renewing consensus in portal hypertension. J Hepatol 2021;Dec30:S0168-8278(21)02299-6. doi: 10.1016/j.jhep.2021.12.022.
- 53. Guevara M, Terra C, Nazar A, Solà E, Fernández J, Pavesi M, et al. Albumin for bacterial infections other than spontaneous bacterial peritonitis in cirrhosis. A randomized, controlled study. J Hepatol 2012;57:759-765.
- 54. Thévenot T, Bureau C, Oberti F, Anty R, Louvet A, Plessier A, et al. Effect of albumin in cirrhotic patients with infection other than spontaneous bacterial peritonitis. A randomized trial. J Hepatol 2015;62:822-830.
- 55. Guevara M, Terra C, Nazar A, Solà E, Fernandez J, Pavesi M, et al. Albumin for bacterial infections other than spontaneous bacterial peritonitis in cirrhosis. A randomized, controlled study. J Hepatol 2013;57:759-65.
- 56. China L, Freemantle N, Forrest E, Kallis Y, Ryder SD, Wright G, et al. A randomized trial of albumin infusions in hospitalized patients with cirrhosis. N Engl J Med 2021;384:808-817.
- 57. Italian Association for the Study of the Liver (AISF). Portal hypertension and ascites: Patientand population-centered Clinical Practice Guidelines by the Italian Association for the Study of the Liver (AISF). Dig Liver Dis 2021;53:1089-1104.
- 58. Campion EW, DeLabry LO, Glynn RJ. The effect of age on serum albumin leveld in healthy males: report from the normative aging study. Gerontol 1988;43:M18-M20
- 59. Wong F, Pappas SC, Curry MP, Reddy KR, Rubin RA, Porayko MK, et al. Terlipressin plus albumin for the treatment of type 1 hepatorenal syndrome. N Engl J Med 2021;384:818-828.
- 60. China L, Skene SS, Shabir Z, Maini A, Sylvestre Y, Bennett K, et al. Administration of albumin solution increases serum levels of albumin in patients with chronic liver failure in a single-arm feasibility trial. Clin Gastroenterol Hepatol 2018;16:748-755.

 61. Mehta G, Jalan R. The "Alter Ego" of albumin in cirrhosis. Hepatology. 2021;74:1734-1736.

- 62. **Starlinger P, Ahn JC**, Mullan A, Gyoeri GP, Pereyra D, Alva-Ruiz R, et al. The addition of Creactive protein and Von Willebrand factor to model for end-stage liver disease-sodium improves prediction of waitlist mortality. Hepatology 2021;74:1533-1545.
- 63. Juanola A, Graupera I, Elia C, Piano S, Solé C, Carol M, et al. Urinary L-FABP is a promising prognostic biomarker of ACLF and mortality in patients with decompensated cirrhosis. J Hepatol 2022;76:107-114.
- 64. Solà E, Kerbert AJ, Verspaget HW, Moreira R, Pose E, Ruiz P, et al. Plasma copeptin as biomarker of disease progression and prognosis in cirrhosis. J Hepatol 2016;65:914-920.
- 65. Macdonald S, Andreola F, Bachtiger P, Amoros A, Pavesi M, Mookerjee R, et al. Cell death markers in patients with cirrhosis and acute decompensation. Hepatology 2018;67:989-1002.

TABLES

ACUTE/SHORT-TERM ALBUMIN TREATMENT	LONG-TERM ALBUMIN TREATMENT
It can last one or more days up to 2 weeks	It lasts at least weeks, usually months or sometimes years
Mostly hospitalized patients* (regular wards/intensive care units)	Outpatients (it can be started during hospitalization)
The goal is to treat or prevent acute complications	The goal is to treat ascites and influence the course of the disease by preventing complications
Effects should occur in hours or days	The effects become manifest usually after 1-2 months of treatment
High daily doses of albumin are infused within a short-time frame	Low doses of albumin are infused distributed over a long-time frame
Safety issues (pulmonary edema) in some cirrhosis complications§	Logistic issues (periodic intravenous infusions)
*patients subjected to large-volume paracentesis albumin in outpatients settings. [§] infections unrelated to bacterial spontaneous perito and worsening or onset of an acute cirrhosis complic	or presenting acute kidney injury can also receiv onitis (ref. #56), hepatorenal syndrome type I (ref. #65 ation requiring hospitalization (ref. #62).

Table 1. Differences between acute/short-term and long-term albumin administration.

Table 2. Main features of the ANSWER and ATTIRE trial. IQR: interquartile range; ITT: intention-to-treat; RCT: randomized clinical trial.

	ANSWER trial	ATTIRE trial
Trial design	Multicenter open-label RCT	Multicenter open-label RCT
Study population	Patients with stable cirrhosis and uncomplicated grade 2 and 3 ascites	Patients with cirrhosis hospitalized for worsening or onset of acute complications
Intervention	40 gr of albumin twice a week for 2 weeks, then 40 gr weekly up to a maximum of 18 months	Targeted to achieve and maintain a serum albumin level >3.0 g/dl from day 3 up to a maximum of 14 days
Primary endpoint	18-month overall survival	Composite of incidence of all-cause infection, renal dysfunction and death between day 3 and 15
ITT population (intervention/control)	431 (218/213)	777 (380/397)
Baseline MELD score (intervention/control) median (IQR)	12 (10-15) / 13 (10-16)	19.5 (15.4-22.9) / 19.5 (15.4-23.4)
Length of treatment in the intervention arm median (IQR)	14.5 (5.0 - 18.0) months	8 (6 - 15) days
Effect on serum albumin concentration	Significant increase in the albumin arm (from 3.1 close to 4 gr/dl)	Significant increase in the albumin arm (from 2.3 to slightly above 3 gr/dl)
Impact on survival	Significantly Increased in the albumin arm	No difference between the two arms
Impact on cirrhosis complications	Significantly reduced incidence in the albumin arm	No difference between the two arms
Risk of pulmonary edema	No	Yes

Table 3. Areas of research on long-term albumin treatment.

Target population stratified according to the expected outcomesBiomarkers of response to treatmentOptimization of doses and frequency of albumin administrationStopping rulesCost-effective analysis in health-care systems world-wide

FIGURE LEGENDS

Figure 1. Properties of the albumin molecule and major changes occurring to the albumin molecule in patients with decompensated cirrhosis. LPS: lypopolisaccharides. PGE₂: Prostaglandin E₂

Figure 2. Potential pathophysiological events antagonized by the oncotic and non-oncotic properties of the albumin molecule in patients with cirrhosis and ascites.

Figure 3. Comparison between the ANSWER and the MACHT trials. *Upper panel:* main features related to albumin treatment. *Medium panel:* changes in the median serum albumin concentration in the ANSWER trial. *Lower panel:* changes in the median serum albumin concentration in the MACHT trial. HA: human albumin; M: midodrine; SMT: standard medical treatment.

Figure 4. *The "filling the gap" hypothesis.* The goal of long-term albumin administration should be to fill the gap existing between the pre-treatment serum albumin concentration and the physiological serum albumin concentration observed in healthy individuals (ref. #62). As the extent of the gap depends mostly by the pre-treatment serum albumin level and the severity of the liver disease, the amount of HA needed to fill the gap (grey arrows) may vary at the individual patient level. The dotted black lines correspond to the lower (3.5 g/dl) and upper (5.0 g/dl) limits of the lab references for defining the normal range of serum albumin concentration measured with standard methods in the daily clinical practice. The red line corresponds to the level of serum albumin concentration during treatment, which has been found associated to optimal outcomes (ref. #45). Modified from ref. #45

ALBUMIN IN HEALTHY INDIVIDUALS

ONCOTIC -

Negative net charge

High molecular weight

High plasma concentration

Regulation of fluid distribution

PROPERTIES -

Most abundant circulating protein

(50-60% of the total proteins)

Reference lab range: 3.5-5.0 g/dl

NON-ONCOTIC

Binding, transport, detoxification

Many endogenous and exogenous compounds including drugs

Antioxidant activity Free radical and metal ion scavenging

Endothelial stabilization

Hemostatic effect

Immune/inflammatory responses modulation

LPS binding, activation of proinflammatory genes, modulation of intracellular redox state, PGE₂ binding

ALBUMIN IN PATIENTS WITH DECOMPENSATED CIRRHOSIS

Structural damages

Reversible and irreversible oxidation Glycation N- and C-terminal truncation Dimerization



Reduced plasma concentration Reduced synthesis Increased catabolism Increased trans-capillary rate

Impaired functions

Reduced antioxidant activity Reduced binding/detoxification capacities Pro-inflammatory effect Others?



	ANSWER	MACHT
Design	Randomized open-label	Randomized double blind placebo-controlled
Length of treatment (median)	14.5 months	63 days
Baseline MELD score (median) SMT/SMT+HA or SMT+M+HA	12/13	17/16
Dosage and timing of albumin administration	40 g twice a week for 2 weeks then 40 g once a week	40 g every 2 weeks (no loading dose)
Impact of treatment on serum albumin concentration	Significant increase (0.7-0-8 g/dl) in the albumin arm	No differences between the 2 groups

ANSWER







Figure 4

ALBUMIN IN HEALTHY INDIVIDUALS

ONCOTIC -

PROPERTIES

Regulation of fluid distribution

Negative net charge High molecular weight High plasma concentration



Most abundant circulating protein (50-60% of the total proteins) Reference lab range: 3.5-5.0 g/dl

NON-ONCOTIC

Binding, transport, detoxification

Many endogenous and exogenous compounds including drugs

Antioxidant activity Free radical and metal ion scavenging

Endothelial stabilization

Hemostatic effect

Immune/inflammatory responses modulation

LPS binding, activation of proinflammatory genes, modulation of intracellular redox state, PGE₂ binding

ALBUMIN IN PATIENTS WITH DECOMPENSATED CIRRHOSIS

Structural damages *Reversible and irreversible oxidation*

Reversible and irreversible oxidation Glycation N- and C-terminal truncation Dimerization



Impaired functions

Reduced antioxidant activity Reduced binding/detoxification capacities Pro-inflammatory effect Others?

Reduced plasma concentration

Reduced synthesis Increased catabolism Increased trans-capillary rate



	ANSWER	MACHT
Design	Randomized open-label	Randomized double blind placebo-controlled
Length of treatment (median)	14.5 months	63 days
Baseline MELD score (median) SMT/SMT+HA or SMT+M+HA	12/13	17/16
Dosage and timing of albumin administration	40 g twice a week for 2 weeks then 40 g once a week	40 g every 2 weeks (no loading dose)
Impact of treatment on serum albumin concentration	Significant increase (0.7-0-8 g/dl) in the albumin arm	No differences between the 2 groups









ICMJE disclosure form

Click here to access/download ICMJE disclosure form Caraceni - ICMJE disclosure form.pdf ICMJE disclosure form

Click here to access/download ICMJE disclosure form Gines - ICMJE disclosure form_ JHEP REVIEW CARACENI.pdf ICMJE disclosure form

Click here to access/download ICMJE disclosure form OBrien - ICMJE disclosure form.pdf